

Weight gain following stroke in younger age (below 70 years) in men and women: challenges and opportunities for prevention and action

HOMER, Catherine, ALLMARK, Peter, BHANBHRO, Sadiq, IBBOTSON, Rachel and TOD, Angela

Available from Sheffield Hallam University Research Archive (SHURA) at:

http://shura.shu.ac.uk/10293/

This document is the author deposited version. You are advised to consult the publisher's version if you wish to cite from it.

Published version

HOMER, Catherine, ALLMARK, Peter, BHANBHRO, Sadiq, IBBOTSON, Rachel and TOD, Angela (2015). Weight gain following stroke in younger age (below 70 years) in men and women: challenges and opportunities for prevention and action. Project Report. Sheffield, UK, Collaboration for Leadership in applied Health Research and Care.

Repository use policy

Copyright © and Moral Rights for the papers on this site are retained by the individual authors and/or other copyright owners. Users may download and/or print one copy of any article(s) in SHURA to facilitate their private study or for non-commercial research. You may not engage in further distribution of the material or use it for any profit-making activities or any commercial gain.





Sheffield Health and Social Care NHS Foundation Trust



Sheffield Teaching Hospitals NHS Foundation Trust

Weight Gain following a stroke in Younger age (below 70) men and women:

Challenges and opportunities for prevention and action

Final Report February 2015

Authors:

Catherine Homer Peter Allmark Sadiq Bhanbhro Rachel Ibbotson Angela Tod

Acknowledgments

We would like to thank all the members of the project advisory group for their advice and expertise. We are also grateful to Amanda Loban and Clare Relton from the Yorkshire Health Study for the data, contacts and help to send out and recruit the survey participants.

Most importantly we are grateful to the participants of the study, who gave their time so generously and who were prepared to share their experiences and views with such honesty.

This report presents independent research by the National Institute for Health Research Collaboration for Leadership in Applied Health Research and Care for South Yorkshire (NIHR CLAHRC SY), a pilot which ended in 2013. Further details about the new NIHR CLAHRC Yorkshire and Humber can be found at <u>www.clahrc-yh.nihr.ac.uk</u>. The views and opinions expressed are those of the authors, and not necessarily those of the NHS, the NIHR or the Department of Health.

Contents

Executive Summary	Page 4
Chapter 1: Introduction and Background	Page 5
Chapter 2: Study aim and research questions	Page 10
Chapter 3: Methods	Page 11
Chapter 4: Results - survey, patient interviews, staff interviews	Page 15
Chapter 5: Discussion and key messages	Page 40
References	Page 42
Appendices	Page 45
Research team	Page 46

Executive Summary

Introduction

This report presents findings from a study to explore the incidence of weight gain in working age adults aged men and women in South Yorkshire. The research was funded by Research Capacity Funds from the National Institute for Health Research, Collaboration for Applied Health Research and Care South Yorkshire (NIHR CLAHRC SY). The project was developed following a research prioritisation event focused on obesity. The event was conducted with staff from a range of stakeholder organisations including health, social care, and voluntary sector.

Methods

Data collection took place between August 2013 and March 2014. A survey was sent to respondents of the South Yorkshire Cohort (Relton et al 2011) who were under 70 years old and had had a stroke. In-depth interviews (n=12) were carried out with patients responding to the survey living in Sheffield and participants from a local volunteer run stroke support group for working aged people. In-depth interviews were also undertaken with professionals (n=18) working across the stroke service pathway in Sheffield. Framework analysis techniques were adopted.

Findings

The research demonstrated that long term weight gain is an issue for some patients following their stroke. Interviews with patients highlighted a range of lifestyle related challenges that people face following a stroke. Reduction in activity levels was reported as a key influence on weight gain.

The health care system and stroke pathway currently provides short-term support to patients to aid recovery and rehabilitation. Longer-term support is harder to access. Existing and new support networks provided by health services, family and friends are essential in recovery and prevention of weight gain. Interviews with staff revealed gaps in their knowledge of their colleagues' roles across the stroke care pathway. Communication between health care professionals across the care pathway was also limited. The presence of patients who gain weight following a stroke was not evident across the entire stroke pathway for example staff in the acute setting were less likely to notice weight gain. Not all staff gave health promotion advice to patients, however all staff thought health promotion advice should be offered to patients, at a time appropriate to the individual and in a manner that will engage the patient in behaviour change. Staff highlighted the need to engage family members in a patient's rehabilitation to prevent long term weight gain.

Key Messages

- The long-term effects of stroke on weight gain are complex. Eating and activity behaviours and social isolation all play a part. Reduction in activity levels was a key influence
- The long-term physiological and psychological impacts of the stroke challenge patients in controlling lifestyle and diet change following their stroke
- The presence and nature of families and social networks before and after a stroke influence weight gain
- Timing of health promotion advice needs be patient centred
- Integrated care pathway can play a part to improve communication between health professionals and patients

Chapter 1: Introduction and background

Introduction

This exploratory study aimed to explore the rehabilitation and weight management experiences of stroke survivors under 70 years of age in the South Yorkshire locality. The purpose was to identify whether weight gain is a problem for people following a stroke, and if so what factors contribute to weight gain.

This is an initial exploratory study that was developed following discussions during coproduction, research prioritisation events run by the obesity theme within Collaboration for Leadership in Applied Health Research and Care, South Yorkshire (CLAHRC SY). Pre-protocol consultation with academics, voluntary sector and health and social care practitioners supported the need for the research.

The report is structured as follows:

- Chapter 1 details the purpose of the study and background
- Chapter 2 includes the study aim and research questions
- Chapter 3 describes the methods employed
- Chapter 4 presents the findings from the survey, patient and staff interviews
- Chapter 5 concludes the report and highlights key messages from the research

Background

Stroke

A stroke occurs when the blood supply to the brain is impaired due to a blood clot or a bleed. A stroke can have a long or short term negative impact on cognition, sight, mobility or communication. 150,000 people have a stroke in the UK each year (Stroke Association, 2012). In 1999, strokes accounted for over 56,000 deaths in England and Wales (Mant et al 2004). In developed countries stroke is the third leading cause of death after heart disease and cancer and is the leading cause of long term disability (Curioni et al, 2009). Approximately a quarter of strokes occur in people of working age, under 65 years (DH 2007). Survival rates following a stroke are increasing, and upwards of 900,000 people in England live with the effects of a stroke (NICE 2010). Whilst the wider economic and societal costs of a stroke are approximately £7 billion per year, £2.8 billion of these costs are picked up by the NHS (Mant et al 2004). The Department of Health's (DH) response to these figures was outlined in a National Stroke Strategy developed in 2007 (DH 2007), which included ambitions for the diagnosis, treatment and management of stroke.

Risk factors for a stroke are split into three categories (Goldstein et al 2001): non modifiable (age, ethnicity, gender, family history); modifiable (includes hypertension, other cardiac disease, diabetes); and, potentially modifiable (obesity, smoking, physical inactivity, drug and alcohol use, hormone replacement therapy and oral contraceptives). Some modifiable risk factors, such as high blood pressure and cholesterol, are linked to lifestyle factors including smoking, low physical activity levels, poor diet and excessive body weight (Kurth et al 2005). Dietary factors, including low levels of fruit and vegetable intake (below 600g per day), and excess salt consumption can impact upon stroke risk. Poor diets are therefore important in primary and secondary prevention of stroke. Adopting or maintaining a healthy lifestyle after a stroke will help to minimise these risk factors.

There is much debate concerning how overweight manifests as a risk factor for stroke. The incidence of stroke is higher in obese people with hypertension or diabetes (Curioni et al 2009). The American Stroke Association guidelines for stroke prevention include being overweight as a risk factor for stroke. The guidelines also advise keeping weight following a stroke to a target body mass index (BMI) of 18.5-25 kg/m2 (Goldstein et al 2006). Kurth et al (2005) highlight a number of papers suggesting obesity as a risk factor for ischemic stroke. However, a Cochrane review, reported that the evidence of an association between obesity and stroke is inconclusive (Curioni et al 2009). Additionally, analysis of clinical data suggests being overweight following a stroke also has benefits for stroke survival (Scherbakov et al 2011). The findings of this research conflict with messages promoting weight management to improve health and reduce the risk of cardiovascular risk. Scherbakov and colleagues (2011) refer to 'the obesity paradox', whereby the impact of being overweight is different in the healthy population to the outcomes of those with chronic cardiovascular disease. They suggest weight management recommendations for patients with chronic illness may need to be different from recommendations made in primary prevention settings, and call for more evidence to be developed from interventional studies.

Modification of health behaviour can be crucial in stroke rehabilitation and in preventing avoidable long term impairment. Rehabilitation is acknowledged as being fundamental to maximising quality of life, independence, and well-being for individuals and their carers following a stroke. However, the success of stroke rehabilitation can be affected by the timing and quality of the information given, as well as access to specialist services. 'Life after a stroke' was a key chapter in the DH National Stroke Strategy. The strategy highlighted the poor quality of and access to rehabilitation services post discharge from hospital. In the first six months, only half of stroke survivors were able to access services; this fell to one fifth in the subsequent six months (DH 2005). Health care and rehabilitation services are weighted towards delivery in the acute phase e.g. focusing on an initial 12 week pathway but only offering limited support thereafter. These services typically target older people and people with complex post-stroke care needs and co-morbidities. As three quarters of younger stroke sufferers want to return to work (Stroke Association, 2006), more evidence is required to understand the experiences of this age group and of people with a less severe stroke. Rehabilitation and recovery specifically targeting to these groups could help to improve quality of life of patients and reduce the costs to the economy and NHS.

The health behaviours and decision making of individuals following a stroke are influenced by existing habits and wider determinants of health such as: income; housing; environment; and, social factors (Niadoo and Wills 2000, Dahlgren and Whitehead 1991). Over the last four years there has been a greater demand for the commissioning of weight management services which support people to improve their diets and increase physical activity levels, whilst also tackling health inequalities. These services could be suitable for some stroke patients. However, local service review suggests they are not utilised by stroke patients and no direct referral from stroke rehabilitation services. Consultation with local expert stakeholders has identified a growing number of patients who are gaining weight as a longer term consequence following a stroke. Data from the South Yorkshire Cohort (SYC) identified that of the 212 people (aged <70 years) who reported having a stroke, 133 are overweight or obese but it is unclear whether this is as a result of the stroke.

Obesity

Obesity is considered a global epidemic and a national and international public health priority and challenge (James 2008, Nanchahal et al 2009, National Obesity Observatory 2010). The 2010 Health Survey for England indicates 42% of men are overweight and 28% are obese, whilst 32% of women are overweight and 26% are obese. Obesity rates in the UK have doubled over the last 25 years and it is predicted to become the most widespread public health concern for adults in the UK affecting 60% of men and 50% of women by 2050 (House of Commons Health Committee 2004, Government Office for Science 2007).

Obesity imposes serious health consequences through increasing the risk of developing type 2 diabetes, some cancers, cardio-vascular disease, and ultimately reduces life expectancy (World Health Organisation 1998, NICE 2006). Furthermore, there are negative social consequences of obesity including discrimination, low self-esteem, social exclusion and loss of working days, all of which impact on the wider economy (NOO 2010). Financial costs to society are estimated to be up to as much as £15.8 billion with UK NHS costs for treating obesity related conditions £4.2 billion (Government Office for Science 2007). Investing in prevention and treatment of obesity could therefore produce massive cost savings which are comparable to that of curing cardiovascular disease or diabetes (Wolfe and Smith 2002).

Obesity is caused by the imbalance of energy in (food consumed through diet) and energy out (energy used by the body to be physically active) (DH 2008). Individual level determinants such as knowledge, practical skills and attitude influence individual responsibility to maintain the balance and much research has concentrated on this area (National Obesity Observatory 2011). However, as above, it is important to take into account the wider social and environmental determinants on health (Niadoo and Wills 2000). Obesity has a number of complex roots (Dahlgren and Whitehead 1991). The Government Office for Science (2007) and DH (2008) suggest these wider determinants often make the healthy choice a hard choice. The relationship between socio economic status and obesity prevalence is an important focus for research (Sobal and Stunkard 1989). Whilst all social groups are affected by obesity, social drivers and the obesogenic environment will play out differently between populations (French et al 2001). Preliminary unpublished analysis of the South Yorkshire Cohort¹ (SYC) data indicates that all social groups experience overweight but obesity is more prevalent in deprived populations. This is reinforced by other studies (McLaren 2007, The Marmot Review 2010). Lower income affects the ability to engage in healthy lifestyle activities for example being able to afford or access healthier foods or access physical activity leisure pursuits (Drewnowski and Specter 2004).

An increased prevalence of obesity is also observed in women and men with fewer educational qualifications (National Obesity Observatory 2010). The National Obesity Observatory (NOO) briefing (2010a) and a systematic review by McLaren (2007) highlights the positive association between increasing levels of deprivation and obesity prevalence in women. Women with low incomes who want to lose weight are less likely to participate in weight loss programmes due to child care responsibilities, location, cost and availability of programmes (French et al 1998). Even where access and cost is not an issue (through free services), people with lower income and educational attainment find it more difficult than more educated counterparts to use combined approaches of increased activity and healthy eating to manage their weight (French et al 1998).

Research development

As the above literature indicates, the associations between stroke, weight and wider social determinants of health are complex. However, there is sufficient evidence to suggest that stroke recovery and secondary prevention would be aided by maintaining a healthy weight. Therefore this study was developed to help establish whether there are risks of weight gain following stroke and why these risks occur. The study will generate insight into the experiences of food and physical activity in a small cohort of people post-stroke who have experienced self-reported weight gain.

¹ The South Yorkshire Cohort is a research project that will follow the lives of 20,000 adults in South Yorkshire over the next 10 years see <u>http://clahrc-sy.nihr.ac.uk/south-yorkshire-cohort.html</u>

User involvement

This proposal was developed and conducted by members of the Obesity theme of NIHR CLAHRC SY in collaboration with the NIHR CLAHRC SY stroke theme and representatives from health, voluntary sector and lay groups.

This project has been developed in response to discussion at a series of coproduction, research prioritisation events involving staff from NHS, local authority, academic and voluntary and community sectors from across South Yorkshire. Here issues were identified that would benefit from exploratory research to develop insight and inform future practices and pathways.

Stroke specialist clinical staff and academics were on an advisory group for the research study. The advisory group provided on-going advice regarding the study design, patient recruitment, study documents, analysis and dissemination.

Chapter 2: Study aim and research question

Study Aim

The study aimed to explore the rehabilitation and weight management experiences of stroke survivors of working age (<70) within the South Yorkshire locality in order to:

- Identify whether weight gain is a problem for some people following a stroke.
- Identify what factors contribute to weight gain.
- Identify appropriate and accessible interventions to prevent weight gain.
- Develop data collection tools for a larger study if required.

Objectives:

- Secondary analysis of the SYC data to explore weight gain prevalence and patterns in people following a stroke (Phase 1)
- Conduct a survey to identify health and lifestyle changes in SYC participants following a stroke and access to post stroke services (Phase 1)
- Conduct interviews with participants and staff to generate in depth understanding of factors influencing weight gain (Phase 2)

Research Questions:

- What are the experiences of weight gain in stroke survivors of younger age and what are the related weight management information and support needs?
- Do current stroke rehabilitation pathway and weight loss/management services provide weight management interventions and information that meet the needs of stroke patients and are accessible to them?

Chapter 3: Methods

Study Design

The study used a mixed method design and included a survey, face to face in-depth interviews and framework analysis methods. The triangulation of the survey and interview data added to the rigour of the findings.

Sample, identification and recruitment

Phase 1 - survey

A population based survey was conducted with participants of the South Yorkshire Cohort (SYC). The cohort includes people who live across South Yorkshire and Derbyshire. The SYC has been granted NHS Research Ethics Committee permissions for the cohort to be used for additional research. For this study, participants of the SYC who consented to be contacted again to take part in future research, were under 70 years of age and had reported having had a stroke were sent the survey (n=193, May 2013). It was not clear from the existing SYC data when the respondent's stroke had occurred or whether the person would still be receiving support under the 12 week pathway of care. It was also unclear whether people were overweight or obese before or after the stroke.

Phase 2 - patient interviews

Semi-structured interviews took place with overweight (BMI = >25) patients who have had a stroke (n=12), and who live in Sheffield. Participants were survey respondents in phase 1 who indicated in their survey response they were interested in the interview participation (n=10) and a participants of a stroke community support group run by stroke survivors (n=2). Purposive sampling (Bowling, 1997; Field & Morse, 1985) was used to ensure the appropriate range of relevant characteristics i.e. age, gender, weight, location, time since their stroke, marital status, socio-economic and work history.

Inclusion criteria for the patient sample included:

- Male / female who have had a stroke
- Living in Sheffield
- Between 18 and 70 years of age
- Able to speak English
- Able to speak (Aphasia friendly)
- Self-reported BMI >25

Table 1- Characteristics of Patients Interviewed

Participant Number	Sex	Age range (years)	Date of First stroke	Weight Change
1	М	50 - 59	2011	Increase 4 stone
2	М	40 - 49	2010	Increase
3	F	30 - 39	2006	Increase 6lbs
4	М	60 - 69	2004	Decrease 22lbs
5	М	70 +	2000	Increase 9lbs
6	F	60 - 69	2010	Decrease 1 stone
7	F	50 - 59	2005	Decrease 1 stone
8	F	50 -59	2007	Increase 10 lbs
9	М	50 - 59	2009	Increase 12lbs
10	F	60 - 69	2001	Increase 21lbs
11	F	60 - 69	2009	Decrease 8lbs
12	F	40 - 49	2010	Decrease 1 stone

Phase 2 - staff interviews

Interviews explored staff perceptions of increases in weight following a stroke and the acceptability and accessibility of services for these patients. Staff and service providers were recruited for interview (n=18). Staff who worked in front line and strategic roles from a variety of health and social care organisations in the voluntary, statutory and community sectors and regularly come in to contact with stroke patients or commission stroke services were contacted by email.

Purposive sampling was carried out following consultation with the project advisory group in order to incorporate staff from a range of services and perspectives.

Table 2 - staff participants

Role	Pathway
Clinical Psychologist	Strategic
Nurse Specialist	Strategic
Speech and Language Therapist	Strategic
Dietician	Acute
Occupational Therapist	Acute
Stroke Consultant	Acute
Dietician	Intermediate Care
Occupational Therapist	Intermediate Care
Physiotherapist	Intermediate Care
Psychologist	Intermediate Care
Sister / Modern Matron	Intermediate Care
Speech and Language Therapist	Intermediate Care
Dietician	Community
Occupational Therapist	Community
Physiotherapist	Community
Stroke Nurse	Community
Sister / Modern Matron	Long Term Rehab
Re-ablement Coordinator	Voluntary Sector

Data analysis

Survey data was analysed using SPSS to generate descriptions of the sample and identify potential relationships between variables.

Interview data was transcribed, anonymised and entered onto password protected computers in Microsoft Office Word. Framework Analysis (Ritchie and Lewis, 2003) methods were used to analyse the data. Framework Analysis involves a systematic process of sifting, charting and sorting the material into key issues and themes allowing the integration of pre-existing themes into the emerging data analysis. NVIVO version 10 was used to store and manage the data.

Strengths and limitations

- The triangulated approach added breadth and rigour to the findings.
- The survey was posted over the summer of 2013 with a covering letter related to the study and a letter from the SYC project manager. SYC, CLAHRC SY and Sheffield Hallam University identifying logos and a unique participant identifier number was included on the survey which linked to the SYC database there a track of who has responded could be provided to the SYC trials team.
- There may have been responder bias in the survey i.e. those who have gained most weight or for which weight isn't a problem may not have responded.

- The sample size was relatively small; although some differences were identified a larger prospective study may strengthen this.
- The study used a pragmatic design but recruited retrospectively, future studies would benefit from a prospective cohort study.
- The study was conducted in one location and would benefit from being expanded to look at other locations.

Chapter 4: Results

4.1 Survey

A copy of the survey can be seen in appendix A. 193 surveys were sent out, 88 responses were received (response rate = 45%). A summary of responses to each question are presented below. Some questions were not fully completed by every participant.



Participants weight before a stroke ranged from 119lb to 252lb whilst after experiencing a stroke their weight ranged from 125lb to 341lb.

There was an increase in weight following a stroke. (P=0.29, difference 5.54lbs, 95% <u>confidence</u> intervals - 0.6lbs - 10.51lbs)

Chart 1 Problems following their stroke



Chart 1 demonstrates the difference in reported problems following a stroke between men and women. Men report more problems with walking. All other reported problems are higher for men but are not statistically significant.

Employment

- Working (full time, part time, paid or unpaid, a carer or housewife/husband) before stroke n=65
- Working now n=29
- Not working due to stroke n=24

The proportion of men and women working prior to having a stroke is broken down fairly evenly with 75% of men in work and 73% of women.

Asking about employment now there is a marked difference with only 26% of men working now and 50% of women working now.

When participants were asked if they are not working now as a result of their stroke 54% of men stated that they were compared to 21% of women.

Chart 2 - Level of activity after stroke



Nearly three quarters of the respondents reported being less active following their stroke. Reasons why activity levels have changed following stroke included:

- "Feelings of fatigue" (Participant ID 61)
- "Difficulty walking and poor balance" (Participant ID 60)
- "Try to exercise more by walking more 2-3 miles a couple of times per week. I have also take up gardening" (Participant ID 6)
- "Lack of interest in any activities" (Participant ID 13)
- *"Retired, more time available for walking, gardening and other activities" (Participant ID 86)*
- "Sadness" (Participant ID 46)
- *"Unfortunately my stroke was caused by too much physical exercise in the gym"* (Participant ID 9)

Of those who had reported that their weight had increased since their stroke, 47% said they were less active. No participants who reported being more active since their stroke has put on weight compared to 0% of respondents who were more active. Those who increased their activity levels were highly likely to lose weight.

Chart 3- Diet following stroke



Nearly half reported their diet as being more healthy and half as about the same following their stroke. Reasons why respondents thought their diets has changed following stroke included:

- "Being less active I know I should eat less, plus I do try to limit some foods I know are not healthy" (Participant ID20)
- *"Can't be bothered to cook. I live on my own no help"* (Participant ID 25)
- "Cutting out salt, sugar. Eating more veg, fruit and fibre" (Participant ID 71)
- *"Direction of spouse!"* (Participant ID 51)
- "Do not want to put on weight" (Participant ID 29)
- "I always had a healthy diet before my stroke" (Participant ID 62)
- *"I eat a lot more healthy, or I try to"* (Participant ID 31)

Of those whose diets are the same or more healthy there was no difference in weight.

Chart 4 - Smoking - before and after stroke



The chart above demonstrates that a third of those who smoked prior to their stroke continued to smoke after their stroke. For those still smoking, the reported numbers of cigarettes smoked each day reduced after the stroke. The rate of smoking prior to the stroke ranged between 5 - 40 cigarettes per day and changed to between 3-20 cigarettes a day after the stroke. In this sample those who stopped smoking since their stroke were more likely to have gained weight since their stroke.

Chart 5 - Units of alcohol before and after stroke



No participant reported drinking more alcohol following their stroke and overall alcohol consumption reduced. Reported reasons for this reduction included:

- "Alcohol not compatible with medication" (Participant ID 84)
- *"Concerted effort to cut down but found this difficult"* (Participant ID 87)
- "Cost" (Participant ID 64)
- *"Do not drink at all now"* (Participant ID 28)
- "Don't go out as much" (Participant ID 65, 73, 43)
- *"More selective, more aware"* (Participant ID 2)
- "My social life has been reduced severely" (Participant ID 62)
- "You only have one life and I want to see my family and grandchildren grow up and a better life for me" (Participant ID 57)

Those respondents who lost weight were more likely to drink less.

Questions were included to ask people where they got advice on diet and physical activity following their stroke. A considerable number of participants (n=30 diet, n=43 physical activity) did not answer or said no to every option on either question. It was assumed that this was because respondents had not received any information on diet and physical activity.

Chart 6 - Diet information from services



Chart 7 - Physical activity information from services



Other sources of information given by participants included:

- "Following my stroke, after regaining use of my limbs I played a large part in the help of others affected by stroke. This put me in contact with various sources of information" (Participant ID 62)
- *"Health services"* (Participant ID 46,55,67,71)
- *"Hospital gave me a big book, stroke handbook. Tells you all you need to know"* (Participant ID 35)
- *"Internet"* (Participant ID 2,9,20,21,27,29,33,72,73,77,82,85)
- "Not on the internet as I cannot use computers etc. because of my stroke" (Participant ID 14)
- "Weightwatchers" (Participant ID 56)

Case studies

The case studies below are summaries from two of the survey participants. Each example highlights the changes that can occur to someone's lifestyle following a stroke. Changes to diet and alcohol are evident; however like many more of the respondents the physiological consequences of the stroke make it difficult to increase physical activity levels. Case study two illustrates that a stroke can be attributed to too much exercise, which can compromise someone's ability to take up exercise again after the stroke.

Box 1, Case Study 1

Case study 1

- Male, aged 49 years
- 2 x strokes, First stroke in2006, subsequent stroke date was not given
- Range of problems since stroke including:
 - o Gained 7st 4lbs
 - Low mobility
 - Not working since stroke
 - Less active since stroke
- Diet is more healthy attending sessions on healthy eating planning
- 200 units of alcohol per week before stroke. Doesn't drink alcohol now
- No support given on diet and exercise
- Self-directed information-internet, family, leaflets

Box 2, Case Study 2

Case study 2

- Male, aged 53 years
- Stroke in 2007
- No problems with activity, but feels depressed and anxious
- Weight loss 1 stone 1lb
- Diet is healthier. He has, cut out biscuits, cakes, chocolate
- Not working now, unrelated to the stroke
- Less active stroke caused by too much activity in the gym
- Diet more healthy checking food and looking out for less saturated fat, sugar and salt
- 20 units of alcohol before stroke, 6 now
- No support given on diet and exercise
- Self-directed information internet, newspapers and supermarkets

3.2 Patient Interviews

In the following section findings are provided in relation to the main themes emerging from the patient interview data.

Table 3 - patient interviews

Themes	Sub themes
Lifestyle	• Weight
	Pre-stroke
	Post-stroke
Networks of support	New support
	Health care support
	Existing support
Impacts of the stroke	Physiological
	Psychological
	Assumptions
Health care	Straight talking
	Navigating
	Integrating
	Closed door
Personal emotion	Treat me
	I'm not old
	Journey to acceptance
	I'm a different person

Lifestyle

Changes to weight and lifestyle pre and post stroke were identified in the data. Some participants reported that their weight gradually increased after their stroke, whereas others talked about losing weight immediately after their stroke. Participants had made lifestyle changes which had impacted on weight loss or weight gain.

Weight: The findings from the twelve patient interviews did not provide a consistent picture of weight gain or loss. Participants were able to say quite precisely the levels of weight change that has occurred after their stroke. This varied between losing weight immediately after the stroke and gaining weight during recovery.

"I'd lost, when I initially, when I came out of hospital I'd lost at least two stone in weight" (Patient P6)

"I've put on around four stone, a fair amount of weight really" (Patient P1)

Participants were also able to talk about their weight and lifestyle at the time they had their stroke and how their weight was influenced by changes in lifestyle brought about by other illnesses.

"I was heavier when I had my stroke, but I have lost weight, I was about 19 stone, but the weight I've lost is not because of the stroke. I have other bladder problems" (Patient P12).

Clear examples were given of lifestyle changes participants had made following their stroke that had contributed to weight gain. Participants who stopped smoking discussed the links with weight gain and how smoking was often replaced with eating:

Those who had active lives prior to their stroke talked about filling their time, with eating.

"I would say sheer boredom. All of a sudden you've got all this time on your hands and you worry and you comfort eat" (Patient P2).

Lifestyle pre-stroke: Health and lifestyle before the stroke was discussed. Some participants reported they were fit and active before their stroke, and some were very active. Others talked about diet and the lengths that they went to, to eat healthily

"So I would think I was in reasonable health, not a marathon runner but in reasonable health, you know. I did smoke, so that might have contributed. But I wasn't taking any statins or anything else for other ailments" (Patient P1).

"I was always one of those people that would get up at half past six every morning regardless. You know, get up, go swimming, go to work, do a full day at work, come home, be doing things. So yeah I was always quite active" (Patient P3).

"I'm fit, healthy. I'm very conscious of my diet, I'm vegetarian, I grow my own veg" (Patient P7).

Lifestyle post-stroke: Participants explained how the stroke has impacted on their lifestyles. Diets, activity levels, alcohol intake and social lives were all affected by the stroke:

"I knew I had a shift in diets because if I were hungry when I were at work I'd just get a Mars Bar or something like that to suffice, a sugar rush or something like that, but now I prefer savoury things to sweet things" (Patient P2)

"I'd say that the social aspect of life was probably the most hit, because I just couldn't be bothered to do anything. I think that being at work during the day was so draining and tiring that when I came home at night I just didn't really fancy doing anything at all" (Patient P3)

"I've cleaned my life up even more because I was a moderate drinker......It ruined my social life when I stopped drinking but that's another issue" (Patient P7)

One patient who noticed they were gaining weight and losing fitness and strength after his stroke spoke specifically about being encouraged to increase his physical activity levels by going to a gym.

"I was very fortunate I was referred to a gym, but it was during that point then you start realising then that actually this, if I don't do something I'm going to lose everything" (Patient P1).

Networks of support

Existing support: Support from wives, husbands and a family was considered to be crucial in recovery. The encouragement and support they provided helped to restore normal activities of living.

"Wives and carers need to be 100% involved in what's going on...... they need to be involved because it's them that's dealing with it 24/7 hours a day" (Patient P2).

"he doesn't always do for me, you know, he lets me do things and, you know. And I'm grateful for that, because it would be very easy just to sit and let him make me a cup of tea or whatever and, but it's not good for you" (Patient P6).

Responsibilities of caring for other family members and pets helped provide the motivation to be active and recover from the stroke.

"And my dog helped...... he's got me going. I've had to get out and do for him." (Patient P6).

Health care support: Patients spoke about a lack of support and information from health services and professionals to help them change their lifestyle after their stroke. This was especially apparent where patients were perceived by health care professionals to be healthy, slim or follow a healthy diet:

"I got no help. I didn't even get a leaflet in hospital or anything like that." (Patient P12).

"I happened to look like the healthiest person they'd had through in a long time. And that got commented on almost every time, oh you're too young to be here, you're too healthy to have had a stroke you know and it's almost like well they gave me, the stroke nurse when she took my history she handed me these A4 papers and then almost pulled them back and handed them me again saying oh you don't really need any of this advice." (Patient P7)

However others commented on their readiness to receive the information on lifestyle and health. They notes that if the timing of information was wrong and people weren't ready or willing to accept information the health care practitioner was wasting their time.

"They probably thought they were flogging a dead horse. I think it were all right, I think diet and that were discussed, but I'm stubborn and I probably wouldn't have taken any notice of diet anyways." (Patient P9)

New support: A participant involved with setting up a local volunteer led stroke group specifically mentioned the lack of support for younger stroke survivors.

"There's nothing in Sheffield, was nothing in Sheffield. They're all like bingo groups, so catering for your older age ladies and gentlemen, sat there playing bingo and one thing or another." (Patient P2).

Feedback about the support provided by this voluntary group was positive as it allowed stroke survivors the opportunity to feel at ease and have fun with others who had suffered from a similar event.

"They're laughing and joking about stupid things, and everybody is the same. Everybody is happy and that's all it takes, a little bit of laughter and a bit of support, and it goes a long way. It really does, that is a lot of the time all that people want." (Patient P2).

".....and Different Strokes are smashing, they really are" (Patient P7)

Voluntary-led support groups can offer long term provision for patients which is something that the NHS did not appear to provide for the stroke survivors in this study. Support groups provided the opportunity to have open communications with other stroke patients to share their experiences rather than learning from literature was important:

"Having open forums where you can go and sit down and talk to people about things, who have lived through it rather than people that have learned about it from a book, makes a lot of difference" (Patient P3)

Impacts of the stroke

Physiological impacts: The extent and type of impacts the stroke had on this sample varied between individuals, as did and the time taken to recover. Participants gave examples of how the physiological effects of the stroke impacted in different ways, ranging from mobility, to vision and balance.

"I couldn't walk around outside really, you know. I couldn't, I could get to really to the gate but I wasn't confident enough about walking over the road or anything like that" (Patient P4) "I lost a lot of my vision. I've got a guide dog; he's in kitchen at the moment. A lot of my left hand side were weak, balance and everything had gone" (Patient P2).

The rate at which the participants recovered from the physiological effects of the stroke varied. Tiredness was something that was reported in a number of interviews:

"But everything else seems to be l've hit that sort of plateau and I think now I can, not exactly concentrate on things, no I suppose it is that, I probably can think well lets address other issues, maybe, you know, the tiredness I guess" (Patient P8)

"My walking is, well I can walk up to about two miles now. I walk with the dog and that's about my maximum, is just two miles and that takes me quite a long time, but I do it. But as I get tired my leg gets more, I find my foot drags more, it drops a bit" (Patient P6)

The physiological impacts of the stroke caused both limited mobility and movement and changes in cognition. All of these could impact on weight gain.

Psychological: Mild cognitive impairment impacted on awareness of food consumption, appetite and ultimately weight gain.

"the memory's not there anymore, how have I put all this weight on? I've not eaten anything, been different, to what I normally eat apart from, four a day doesn't mean four packets of bloomin' biscuits a day" (Patient P7)

Loss of memory frustrated participants and embarrassed them in front of their friends and families:

"Yeah, my memory is absolutely appalling. I went to shop for one thing and I come back, what did I come down for? And I buy all sorts of crap, and I'll come back and she'll say did you get bread? Oh hell, back down the shop for some bread, the one thing that I went for that I just totally forgot about it." (Patient P2)

The stroke also triggered strong emotions. Distress and avoidance of certain situations became normal. Whilst this improved for some during the recovery process, other participants talked about having to learn to be in the company of others again, limiting their social activity:

"Initially I found I was very emotional in certain situations. I'm a lot better with that now, but certainly my brain gets tired. I don't like rooms with a lot of people in; the noise is too much for me sometimes. But I cope. I perhaps possibly don't put myself in the situations that I would find difficult if I can avoid them." (Patient P9)

Health care

Straight talking: Participants thought that medical staff did not always give them enough detail and information about their stroke and potential rate of recovery. They requested that health practitioners were more honest and realistic about the time and extent of recovery possible:

"I do feel that when it first happened, when I first went to the stroke assessment clinic, I think at that point it would have been a lot nicer for someone to sit down properly and explain what was going on and why it might have happened, which I think is the one thing that was lacking at the time, of someone just having half-an-hour explaining it properly" (Patient, P3).

"I think they ought to be able to talk to the patients more about how it might affect them mentally" (Patient P8).

Furthermore, the ways in which health staff communicated with patients was considered important. It was necessary to speak in a language that patients can understand and refraining from using medical terminology. This helped participants to process and understand what had happened and what was happening to them:

"He was just straight. He spoke to me on my level, rather than trying to be medical about everything" (Patient P9)

"He [consultant] sat me down and explained everything to me so I wouldn't get frightened again" (Patient 12)

Navigating: Participants reported problems navigating NHS services and stroke support. One patient's family had to seek additional support. This meant they then went from receiving little help to being overwhelmed with the number of people and organisations that got involved:

"my wife went down to the doctors and said, look, my husband had a stroke seven week ago, nobody has been round to house and things are getting really bad, to the point of despair. So he got in contact with, it were the CICS team then. I can't remember what they were called, anyway the physio teams around Sheffield, and then they started coming, the physio team, the occupation therapy and then the Royal Society of Blind...... So all of sudden from nothing, you've got hundreds of people just coming all at once, and that's a hard thing to try and deal with. Your house is your own and then it's not" (Patient, P2)

Closed door: At the time the participants had their stroke patients were supported by the community stroke team for 12 weeks. After this they were discharged, referred back to their GP or onto further stroke services if clinically appropriate. For those participants who weren't referred for further support this was a worrying time and patients reported feeling isolated and unsure where to turn for further support:

"We've done what we can (referring to Community Stroke Team)....that's all we can do for you now....I'm thinking what happens next? Is it down to me? is there something out there? but there is nothing out there" (Patient P1).

Personal emotion

Treat me: Being treated as an individual was important to the participants in their recovery process. Setting personal goals that would aid rehabilitation and re-assemble their previous life was considered valuable. Personal goals were perceived more important than some service goals that seemed like a tick box exercise to enable services to discharge them:

"I didn't feel that I was look [looked] at as an individual...... There was the thing in the kitchen, could I make a cup of tea by myself. Yeah okay I can do that. I didn't pour water over my hand or anything else like that I was able to do that brilliant. So I was safe and I get that part of it......You know, I am this age, I want to be getting to there, how are you going to help me get to that point, what can I do that's going to make that, what can I do now that's going to get me up there. And there was none of that really. There was nothing that really helped that they did that was really going to help me, other than routine". (Patient P1)

I'm not old: Age was a theme to emerge from across all the interviews. From the acute setting through to community based services, participants referred to experiences where they had felt they were treated like an old person, not someone of working age. In the acute setting many patients around them were old, and they felt out of place.

"I were put with a load of old age people, and no disrespect to anybody that's of the older years, I couldn't get any rest because there were buzzers going off all the time" (Patient P2).

"You've got people on same ward, like 65, 70-year-old, fair enough, things happen, but at 45 it's like hang on a minute, I shouldn't be here" (Patient P9).

Out-patients services were also considered to be set up for the elderly population, rather than younger people who had experienced a stroke:

"The health, because you had to keep going back down to [Name of Hospital] Hospital and where do you go? You go into the elderly care centre, which is not very nice to go in there, and you sit down and you look around. All sit around and you find yourself surrounded by old, old people, and I'm not being nasty to them, well I hope I'm not, you know, I just didn't feel like I was in the right place" (Patient P1)

Community services didn't always consider the needs of younger people who wanted to go out with their families:

"And I sort of said look, I'm a young person, I've got young children, I mean I've got things to do. And I was still trying to do them. I felt that you get treated like a 70-year-old who's had a stroke. And obviously I know that young people, under a certain age, under 70 is a very

small minority, most people are over 70 and you just felt like this is the way they treat then, you know, you're an old person you'll be sat at home doing nothing, but actually no I've got things I want to do. I still want to go shopping, I still want to go and visit people, I still want to get back into work and things, so I was trying to do things and, you know, I'm just not on their beck and call here, you know, it's not as easy as that" (Patient, P1).

Journey to acceptance: Accepting the stroke has occurred and the physiological, psychological and emotional impacts takes time and was very much an individual process. Some participants were quite blunt about how they had to accept what had happened.

"You know, well it is, you live with it, well you've got no bloody option really have you?" (Patient P4).

Clear examples of the journey to acceptance were given. The quotes in the case study 3 below from participant 2 demonstrate this:

Box 3, Case Study 3

Immediately after the stroke "Was I quite poorly? I don't think I were. My wife would disagree, and I suppose everybody else would disagree. Dealing with it within my head, I'm a very, am I still an alpha male? Yes, I am. I was a very alpha male, so as far as pride and everything else that went with it, it messed about with my head big style, it really did."

Fighting back "it's that battle of wills within yourself; it's that, that's hard. For me personally, I can't speak for anybody else, but what don't make you will break you, and it made me, it didn't break me so I'm lucky on that aspect.

Reflections and the future "The hardest part of that 1,000 mile journey is a first step because it's not 1,000 miles anymore, and each and every step that you make after that it gets closer, and it gets easier, but the hardest step is the first step. And after that you want it more, so you get more hungry or the pace starts getting faster and faster, yeah, definitely.

There were many examples of people fighting to get on with things and with internal motivation to improve. Some participants reflected that their stroke could have been much worse and they had got away lightly compared to others that they know.

"But I think some people lose their drive to do things. And I'm maybe lucky that I've, from the very first day that, when I knew I'd managed a night and I was still alive, with my heart and the stroke I was determined I was going to make the best of it I could do and I've stuck with that really all the way. It's hard, it's very hard" (Patient P6).

I'm a different person: Participants discussed changes to their physical and psychological self. Their ability to work and care for others and provide for the family, had changed. Some now had to rely on others to eat, dress and socialise. This meant that participants felt that they were a different person

as a result of their stroke. In some cases this was judged to be for the better as the recovery had provided a time to reflect and readdress where their life was heading. In other circumstances participants found it harder to accept the differences. This not only meant that they themselves needed to learn who they were and what they could do but also that their family and friends had to deal with the changes and this 'new person':

"So everything in your life is suddenly changed and it's almost like you've been off, sort of tilted of centre and. You're still running your life, you're still trying to do your life, but you're just suddenly you're not able to do the things you could do. And it's simple things, you know, your writing's affected, your speech is affected. If you're trying to do your normal routine stuff, you know, banking and talking to people on the telephone, it's all affected by it" (Patient, P1)

In summary the patient interviews build on insight generated from the survey revealed that some people did struggle with weight gain following their stroke. Experience of weight varied following the stroke and some gained weight. Stopping smoking, boredom, reduced activity and the psychological impact of the stroke contributed to weight gain. For those who gained weight the stroke increased social isolation and impacted on other health behaviours e.g. alcohol consumption. The nature and timing of information to prevent weight loss and maintain a healthy weight varied across participants.

3.3 Staff Interviews

In the following section findings are provided in relation to the main themes emerging from the staff interview data.

Table 4 - Staff Interviews

Themes	Sub themes
Control	• Loss
	 Physical impairment
	• Communication
	o Cognition
	Socio-economic status
Support	Communication
	 Intra professional organisational uncertainty
	 HCP responsibility
	 Conflict
	Gaps in knowledge
	 Monitoring of weight gain
	 Knowledge of patients
	 Knowledge of HCP and wider services
	Type of information
	 Health promotion advice for patient and family
	 Bespoke approach
	Services
	 Voluntary sector
	 Stroke care pathway and transition
	 Service cuts
	 Access
Weight gain	Impact
	Influence
	 Food providers
	 Social isolation
	 Psychological effects
	 Physiological effects
	 Lifestyle before stroke
	o Age
	Prevalence
	• Weight loss
	 Timing of onset
	 Does exist

Control

Staff described the loss of control people have following a stroke. The physical impairments and problems with communication and cognition all have major impacts on the extent of recovery and time taken for patients to recover. The extent to this loss of control is far-reaching and staff viewed it as a critical influence on the potential for future weight gain.

Physical impairment: The impact of changes in physical ability following a stroke was considerable. Loss of mobility and vision led to social isolation. Even in the home patients may well need to adapt their environment or have aids to help them get around. Staff reported that patients who lose control over aspects of their lives take time to accept this, especially in those people who were not chronically ill prior to their stroke. The combination of loss of control, mobility and physical impairment was thought to be an influencing factor on weight gain.

"They've lost their independence; they've lost the freedom to go out. Most of these people were able-bodied beforehand and have become suddenly very disabled, so they've got to come to terms with their new disability" (Staff, P1)

"For the majority of people if they haven't got any disabling illnesses before it comes as a big shock having a stroke. I mean for working age people it stops you driving, you're not allowed to drive, your environment, your home environment may need changing, you may become incontinent, you're not able to walk about" (Staff, P4)

Communication: In relation to eating and drinking, losing the ability to communicate with friends, family and carers was considered to have a long term impact on weight gain. Staff highlighted issues in those patients who can't express to others what they want to eat, therefore having to rely on food being provided.

"She wasn't able to express clearly as well around what she wanted to eat actually, so those decisions were made by family or carers" (Staff, P2).

Cognition: Like the patient interviewees, the staff reported how cognitive impairments can further influence the chances of gaining weight. For example through losing the ability to control portion sizes, decide what foods to have or whether to take exercise, and the increased reliance on others as food providers:

"Stroke affects appetite, so you can actually get, because of the brain damage in certain areas of the brain, can cause, it's a bit like losing inhibitions. You lose the ability to decide what's the right quantities, so it requires somebody else to decide for you" (Staff. P1)

Staff also reported that patients were not always aware that they were increasing their weight. Examples were given of people whose lifestyle changed due to cognitive impairment and because of this were gaining weight: "I can think of one patient where that has been an issue, where the portion sizes haven't changed in relation to the level of physical output, so she did end up putting on weight. She had cognitive impairment so it meant that she wasn't in a position to take on board and rationalise and manipulate the information to be able to understand and express a wish to have smaller portions, even though she was aware she was putting on weight. (Staff, P2)

Participants suggested that for those patients where cognitive impairment severely restricts lifestyle change, there is no real point trying to encourage patients to change.

"...plenty of patients where it's impossible to expect lifestyle change if there's significant cognitive impairment" (Staff, P14).

Socio-economic status: Staff participants described how socio-economic status was linked to stroke. This was both as a causal factor of poor lifestyles prior to the stroke and also by hindering the chance of engaging in positive lifestyle choices following a stroke:

"Where strokes tend to happen, if you look at some of the areas of Sheffield where we're looking at the poor socioeconomic areas, where making healthy choices is not something people have been doing anyway...... But certainly in some areas you get a lot more of the younger strokes that are overweight already, that tend to carry on with their unhealthy lifestyles. And that's part of almost a community thing" (Staff, P1)

Changing behaviours was considered more difficult by staff when patients lived in areas of increased deprivation, where income and access to quality food was low:

"Those innocent veg pots, they seem quite good. Yeah, I suppose they're better than having fat, but a lot of people can't afford. Yeah, they're much dearer. The things, they're a lot more expensive and a lot of people go perhaps to Iceland or things and the fat content in things is really high" (Staff, P5)

Support

Health Care Professional (HCP) responsibility: There was evidence of conflict between staff groups regarding the times at which patients should be given lifestyle information. Some thought it should start early in acute settings. However staff in the acute settings reported that they lacked the skills and time required.

"well I think that [health promotion advice] starts very early after a stroke. I think that possibly starts when people are still in hospital" (Staff, P10).

"In an acute setting I don't think it's necessarily our role to give them health promotion advice" (Staff, P9

"I guess I don't need to have an in depth understanding of the pathway. I just need to deal with the patients on my ward" (Staff P9).

As a result of the relatively short (12 week) stroke pathway, services provided by the voluntary sector or those commissioned outside of the NHS could provide a vital support and rehabilitation opportunity to stroke patients. However staff were reluctant to refer to organisations and services outside of the NHS for fear of any comeback:

"It wouldn't be within our remit to direct the person to an individual thing. Other than general advice, you know, we might all give, but we wouldn't specifically refer them to another service without going back through the GP" (Staff, P6)

"We are all pushed to encourage people to think about voluntary sector and all that, but when it comes to it I don't think the majority of staff don't, partly because, this is going to sound awful, partly because, if we lead somebody down that route and they get involved in something, we almost hold responsibility for it" (Staff, P13)

Monitoring of weight gain: Staff participants were unclear about the extent to which weight gain was an issue for patients after a stroke. Weight was not routinely monitored across all departments, nor was it monitored long term. Various practical issues related to weighing patients in the community, and with patients with limited mobility were highlighted. However some staff participants were clear that weight gain was an issue with many patients, especially those that they come into contact with patients one to two years following their stroke. Participants said there was limited access to specialist services to refer to if they do highlight an concern about a patient's weight:

"We didn't really do BMIs on everybody until fairly recently in the last year or two, there is more that culture. And I systematically now whenever I see patients in clinic will send out a fact sheet on healthy eating; I will be more proactive in referring for dietetic support, even though there isn't an awful lot there in the community, there's always a waiting list and that, and then to focus a lot more on lifestyle" (Staff, P5)

Type of information

Health promotion advice for patient and family: Many staff offered health promotion advice to their patients but they were not always confident of how this would impact on behaviour change. Even where patients took part in practical physical activity sessions staff were not confident patients maintained this:

"I'll leave brochures and leaflets and smoking cessation stuff with them, knowing full well they're not going to do anything with it. But at least I can, I suppose in a sense I can tick my box to say they've got the information. And some of it's about cognitive impairment, you know, so it isn't easy" (Staff, P1)

"The physical activity that we encourage here, and obviously we can only, again we can give people advice, but whether they follow what they've been told why they're here is another thing" (Staff, P4)

Staff spoke of different ways they give the information about weight, lifestyle and stopping smoking and examples were given of using pictures of blood vessels, poor diets and smoking.

"I developed a way of approaching it so it's not focussing on the weight; I focus it on the stroke, so talking about its good to have a healthy diet to think about reducing the risk of stroke" (Staff, P8)

Bespoke approach: The timing of lifestyle advice and the manner it is given to patients was considered to be crucial. HCP's stressed the individual nature of acceptance and how people respond differently to the stroke:

"Some patients will see this is the warning shot, this is it, I can't believe I've been doing that, I'm going to change my ways, I'm going to do what you say. Some patients will need to hear it again and again and again. Some patients won't change what they do, irrespective of how many times you tell them" (Staff, P17)

"I'll go back a bit later and say let me come back and talk to you at another time, because timing is critical very often, with all of these things. And lifestyle changes you need hit people at the right moment. So I may come back later and talk to them again about it" (Staff, P1)

Services

Stroke care pathway and transition: The timing and length of the pathway was a theme in many of the interviews. There was concern that the length of the care pathway had reduced in recent years.

"In the past we would have had people in for longer" (Staff, P9).

Staff were sad that the time frame that they had to support patients in their recovery was so short. They saw that patients were being discharged even when staff knew patients would benefit in the long term from more clinical input:

"In an ideal world I would like there to be, I mean you're going to probably hear this from everybody, a long term stroke team in Sheffield... where patients can be monitored over years really, and have access to services that can really, and I'm talking as an occupational therapist, that can really look at overall occupational balance and function, and lifestyle as a whole, because obviously weight fluctuation, weight gain is complex. I mean it would be great as well if there were more opportunities for people to be able to access alternative environments easily with support" (Staff, P2)

Access: Staff felt that access to dietetics, community based weight management or physical activity services was poor. Where services existed staff spoke about long waiting lists.

"There is a service for people who are very overweight, but that's very underfunded and people have to wait months and months to access that service. So even where people have a long established identified problem it's not fantastic" (Staff, P3).

Transport and physical access to community based services was problematic for stroke patients. They described how, even where community transport existed patients suffering from poor mobility and cognition found it difficult to navigate the transport booking systems. The capacity of staff to provide the additional support to help patients access services to help their recovery was limited. Home based services were much easier to organise.

"The people who really need our services are the ones that can't access it, or those services rather, because they can't get out and about. Smoking cessation do a home visiting service which is great, but for things like exercise and activity stuff......." (Staff, P1).

Service cuts: There was a general feeling of reductions in the scale of community based schemes which promote exercise and physical activity:

"Sheffield used to be quite well set up in that people could easily access gyms and have, people that worked in the gyms could mentor them with a programme, and they were referred by their GPs and that sort of thing, but it seems to be a lot more problematic to access that sort of thing now" (Staff, P3)

Financial constraints and cuts to services were also thought to be increasingly problematic, particularly for those that are socially isolated

"There are less and less services. We used to rely heavily on luncheon clubs and respite care for the people that come in and have a period of time, none of that's available now" (Staff, P4)

Weight gain

Influence - Food providers: Following a stroke many patients cannot prepare their own meals and rely on family members, friends or carers to provide their food. Staff thought that this may influence whether a patient gains weight or not. Carers have little time to spend cooking fresh healthy food and rely on quick microwave meals which staff considered to be unhealthy, highly calorific and often laden with salt. Participants explained that some families have a limited understanding of healthy diets and portion sizes. 'Treats' to make stroke patients feel better were common place:

"when patients go out into the community, if they're not in a position to be independent with their meals and have control over shopping, which is the majority, then it is in the hands of either agencies or relatives. And what I've seen is relatives, I mean I think it's a way of showing affection, it can be a way of showing affection, buying food that their relative who's had the stroke likes is one factor" (Staff, P2)

Social isolation: Stroke patients often become socially isolated because of the residual effects of the stroke. This impacted upon their diet, activity levels and vulnerability to weight gain:

"I saw a 48 year old, he hadn't got a very good social network, hadn't got a lot of support, big chap who was putting on weight all the time. He just couldn't prepare food, we couldn't get social services support enough for him, and it was just really difficult about how you could help him really without me going there every day and cooking for him" (Staff, P5)

Patients may spend much more of their day at home and often resort to food as a comfort.

"But they don't have anything in life, they're very lonely, they're isolated, so eating is the thing that keeps them going really. And it's difficult when I see those people to say well don't do that, because there is nothing else to replace that" (Staff, P1).

Psychological and physiological effects: Depression is strongly associated with stroke; staff participants in this study were able to give examples of how patients' mental health was affected. Impacts included using alcohol intake to make them feel better which may also impact on long term weight gain.

"In quite a number of cases increased their alcohol intake after the stroke because they've been fed up and depressed" (Staff, P5)

Along with changes to diets, reduced mobility and activity is a common consequence of strokes:

"Less activity, the vast majority of patients are spending a lot of time either in bed or sitting in a chair; they're not spending huge amounts of the day mobilising. They're the sort of patients that tend to put the weight on" (Staff, P3)

Weight gain was thought to be more apparent for those people who were more active prior to their stroke. Furthermore staff thought that people who were more active often continued to eat the same portion sizes which contributed to their weight gain.

"the people I see that I've noted gain weight have been the people who have been quite mobile prior, and then the stroke affected their mobility in some way" (Staff, P13).

Prevalence

Weight loss: Not all staff across the stroke pathway considered weight gain to be an issue for stroke patients. Those staff working at the acute end of the pathway were more focussed on weight loss, and rarely came across patients were weight gain was an issue to manage.

"we worry about people that are underweight, who are nutritionally malnourished" (Staff, P4)

"[weight gain] may become more of an issue later on but we don't see it" (Staff, P7).

Whilst a few staff did not see weight gain as a concern the majority agreed that people following a stroke were vulnerable to weight gain. For those who did not recognise weight gain as a problem this may reflect the position at which they see a patient within the stroke care pathway. Staff described various physical and psychological impacts of the stroke which from their perspective can contribute to long term weight gain. They felt restricted in responding to information and support needs regarding weight management because of the constraints of the current time frames of the commissioned stroke pathway and the focus on the acute phase.

Chapter 5: Discussion

This study aimed to identify whether weight gain is a problem for some people following a stroke. The findings indicate this is so. The study also provides insight into the physical, psychological and social factors in some people following a stroke from the perspective of patients and staff. The current stroke service is restricted in its ability to deliver timely and individualised information, advice and support to prevent long term weight gain and promote weight management.

Key messages to emerge from the findings are now summarised followed by the main implications for practice.

Key messages:

Weight Gain

- The long-term effect of stroke on weight gain is a complex issue influenced by a range of pre -existing environmental factors and behaviours and the physiological and psychological effects of the stroke.
- Eating and activity behaviours and social isolation all play a part on weight gain, with the reduction in mobility and levels of activity being a key influence.
- Patients are challenged in controlling their diet following their stroke.
- Families and social networks influence weight gain.
- Long term follow up support would provide additional opportunities to influence the diets of patients and carers to reduce weight gain and risk factors for future chronic conditions.
- Social isolation and depression are a major factor in weight gain.
- Loss of control in cognition and communication impacts ability to manage diet and weight.
- Patients' socio-economic status impacts pre and post lifestyle choice.

Stroke pathway:

- The timing and type of health promotion advice needs to take a patient centred approach.
- The restriction of the pathway of support to the twelve weeks post stroke for most patients impacts the quality, appropriateness and acceptability of advice given to patients and carers.
- Care and information provided to stroke patients is restricted by pathways, availability of specialist support and current funding cuts to community based services.
- Fragmented communication between health care professionals across the pathway limits opportunities to address the risks and realities of weight gain.
- Ambiguity arises for patients because weight gain following a stroke can be both a sign of progress and of a problem.
- It remains unclear how common weight gain is as it is not a priority, monitored routinely or monitored long term.
- Knowledge and ability of wider support services hinders patients' recovery.
- Limited knowledge and availability of activity outside of NHS provision.
- Physiological impacts of stroke restrict possible engagement in community activity.
- Constrained knowledge of the whole stroke pathway roles and responsibilities.
- Restricted support due to length of patient contact.

Implications for Practice

- The true extent of stroke related weight gain is unknown because services do not monitor or document weight across health care settings. Therefore a routine documentation of weight is required for people following a stroke as people move from acute, rehabilitation and survivorship phases of their care delivered in different settings.
- Health care professionals in different settings need to be mindful of the risk that people are susceptible to long term weight gain following stroke.
- Whether in acute, intermediate, rehab or primary care settings systems should be in place to identify opportunities for advice and support regarding diet and physical activity e.g. Making Every Contact Count (MECC).
- Information and support needs to be delivered in a timely manner and in an acceptable and accessible way to individual patients and their families.
- More work is required to ensure that messages regarding weight, diet and physical activity are consistent across settings.
- The study raises concerns regarding staff knowledge and awareness and the available systems to ensure referral occurs to weight management services.

Implications for research

- A large cohort study would provide population data to explore the prevalence and causes of weight gain following a stroke.
- The data collection tools developed in this study could be used in further research to explore the issue of weight gain.
- The current study was restricted in its patient sample selection for expediency it recruited from an existing cohort study. There would be merit in conducting a l prospective longitudinal study mapping experiences of weight, diet and physical activity in people following people from patients from acute care to two years post stroke.

References

Bowling A. (1997) Research Methods in Health: investigating health and health service. WB Saunders Co. Philadelphia.

Curioni, C., Andre, C., Veras, R. (2009). Weight reduction for primary orevention of stroke in adults with overweight or obesity (Review). The Cochrane Collaboration.

Dahlgren, M., and Whitehead, M. (1991). Policies and strategies to promote social equity in health. Institute for Future Studies, Stockholm

Department of Health, 2005, The National Service Framework for Long-term Conditions, London, DH

Department of Health (2007). National Stroke Strategy. London. Last accessed on 16th January at: <u>http://www.dh.gov.uk/prod_consum_dh/groups/dh_digitalassets/documents/digitalasset/dh_0810</u> 59.pdf

Department of Health (2008). Healthy Weight, Healthy Lives: A Cross Government Strategy for England. [online]. The Stationary Office. London. Last accessed on 17th August 2011 at: <u>http://webarchive.nationalarchives.gov.uk/20100407220245/http://www.dh.gov.uk/prod_consum_</u> <u>dh/groups/dh_digitalassets/documents/digitalasset/dh_084024.pdf</u>

Drewnowski, A., and Specter, S. (2004). Poverty and Obesity: the Role of Energy Density and Energy Costs. American Journal of Clinical Nutrition. 79;6-16.

Field PA. Morse JM. (1985) <u>N</u>ursing Research: the application of qualitative approaches. Chapman & Hall. London

French, S, A., Story, M., Jeffery, R, W. (2001). Environmental influences on eating and physical activity. Annual Review of Public Health. 22;309-35. Last accessed on 21st August 2011 at http://web.ebscohost.com.lcproxy.shu.ac.uk/ehost/pdfviewer?sid=45cf7c95-3c27-4173-afee-f4e2016f4af6%40sessionmgr110&vid=2&hid=119

FOOD Trial Collaboration. (2003). Poor nutritional status on admission predicts poor outcomes after stroke. 34:1450-1456.

Goldstein, L, B., Adams, R., Becker, K., Furberg, C, D., Gorelick, P, B., Hademones, G. (2001). AHA Scientific Statement: primary prevention stroke: a statement for healthcare professionals from the Stroke Council of the American Heart Association. Stroke; 32:280-99.

Goldsteain, L,B., Adams, R., Albers, G., Alberts, M,J., Benavente, O., Furie, K., et al. (2006) Primary prevention of ischemic stroke: a guideline from the American Heart Association/American Stroke Association Stroke Council. Circulation. 113:873-923.

Government Office for Science (2007) Tackling Obesities: Future Choices. London: Department for Innovation. Last accessed on 17th August 2011 at <u>http://www.bis.gov.uk/assets/bispartners/foresight/docs/obesity/17.pdf</u>

Hackett ML, Yapa C, Parag V and Anderson CS, 2005, 'Frequency of depression after stroke:a systematic review of observational studies', Stroke 36 (6), 1330–40

Health Survey for England (2010): Trend Tables. The NHS Information Centre, 2001. Last Accessed on 21st January 2013 at <u>http://www.ic.nhs.uk/pubs/hse10trends</u>

House of Commons Health Committee (2004) Obesity: Third Report of Session 2003-04. London: The Stationary Office. Last accessed on 24th June 2011 at http://www.communities.gov.uk/documents/communities/pdf/733520.pdf

James, W, P, T. (2008). The epidemiology of obesity: the size of the problem. Journal of International Medicine. 263(4); 336-352.

Kurth, T., Gaziano, M., Rexrode, K,M., Kase,C,S., Cook,N,R., Manson,J,E., Buring,J,E. (2005). Prospective study of Body Mass Index and Risk of Stroke in Apparently Healthy Women. Circulation. 111: 1992-1998. Last accessed on 16th January 2013 at http://circ.ahajournals.org/content/111/15/1992.full.pdf+html

Mant J, Wade DT, Winner S (2004) Health care needs assessment: stroke. In: Stevens A, Raftery J, Mant J et al., editors, Health care needs assessment: the epidemiologically based needs assessment reviews, First series, 2nd edition. Oxford: Radcliffe Medical Press, p141–244.

McLaren, L. (2007). Socioeconomic Status and Obesity. Epidemiologic Reviews. 29:29-48.

Nanchahal, K., Townsend, J., Letley, L., Haslam, D., Wellings, K., Haines, A. (2009). Weightmanagement interventions in primary care: a pilot randomised control trial. British Journal of General Practice. 59:349-355.

National Obesity Observatory. The economic Burden of Obesity. Association of Public Health Observatories. October 2010 (b). Last accessed on 17th August 2011 at <u>http://www.noo.org.uk/uploads/doc/vid_8575_Burdenofobesity151110MG.pdf</u>

National Obesity Observatory. Data Sources: environmental influences on physical activity and diet. NOO Data Briefing. Association of Public Health Observatories. March 2011. Last accessed on 17th August 2011 at

http://www.noo.org.uk/uploads/doc/vid_10418_Environmental%20%20data%20sources%20FINAL_editedformatted_%20MG%20100311.pdf

NHS Information Centre (2009) Health Survey for England 2008. Joint Health Surveys Unit. Last accessed on 17th August 2011 at

http://www.ic.nhs.uk/webfiles/publications/HSE/HSE08trends/Health_Survey_for_england_trend_t ables_2008.pdf

Niadoo, J. and Wills, J. (2000) Health promotion: foundations for practice, Edinburgh: Balliere Tindall

National Institute for Health and Clinical Excellence. (2006) Obesity: the prevention, identification, assessment and management of overweight and obesity in adults and children. London. Department of Health. Last accessed on 7th February 2011 at: <u>www.nice.org.uk/guidance/CG43</u>

National Institute for Health and Clinical Excellence. (2010) Cost impact and commissioning assessment: quality standard for stroke. Last accessed on 16th January 2013 at:

http://www.nice.org.uk/media/87D/33/StrokeCostingCommissioningAssessment.pdf

Relton et al. South Yorkshire Cohort: a 'cohort trials facility' study of health and weight - Protocol for the recruitment phase. BMC Public Health 2011, 11:640 <u>http://www.biomedcentral.com/1471-</u>2458/11/640

Ritchie, J., Lewis, J. (2003) Qualitative research practice. London. Sage.

Ritchie, J., Spencer, L. (1994) Qualitative Data Analysis for Applied Policy Research In: Bryman A, Burgess G. (eds) Analysing Qualitative Data London. Routeledge. pp 173-194

Scherbakov, S, N., Dirnagl, U., Doehner, W. (2011). Body weight after stroke: Lessons from the obesity paradox. Stroke. 42:3646-3650.Last accessed on 16th January 2013 at <u>http://stroke.ahajournals.org/content/42/12/3646.full.pdf+htm</u> I

Sobal, J., and Stunkard, A, J. (1989). Socioeconomic Status and Obesity: A Review of the Literature. Psychological Bulletin. 105(2): 260-275.

Stroke Association and Different Strokes (2006). Getting back to work after stroke.

Stroke Association (2012). Struggling to Recover. Life After Stroke Campaign. London.

The Marmot Review (2010). Fair Society Healthy Lives. The Marmot Review Team. Last accessed on 16th August 2011 at

http://www.marmotreview.org/AssetLibrary/pdfs/Reports/FairSocietyHealthyLives.pdf

Wolfe, B,. and Smith ,J,E. (2002). Different strokes for different folks: why overweight men do not seek weight loss treatment. Eating Disorders. 10;2:115-124.

World Health Organisation (1998). Obesity: Preventing and Managing the Global Epidemic. Report of the WHO Consultation on Obesity. Geneva: WHO

World Health Organization (2002). The World Health Report 2002: Reducing Risks, Promoting Healthy Life.

Appendices

Appendix 1 - Survey

She Hal Uni	ffield lam versity	Centre and So Resear	for He cial Ca ch	alth re	Γ	For office use only
CLAHRC	for South Yorkshire	Nation]] nal Institu lealth Res	te for earch		
ivit_1	South	Yorkshire Col	nort - Stroke	Questi	onnaire	
We are o you com your life given ou	ontacting you follo pleted. We would style. This will help t to people followi	owing on from the to ask you for the to ask you for to understand a stroke.	te South York or some inforr nd more abou	shire Co nation al t the hea	hort Heal bout the s ith prom	th Questionnair stroke you had a otion informatio
The que Once yo Sheffield	stionnaire will take u have completed Hallam University	5-10 minutes to the questionnair in the pre-paid	e complete. Ple e please send envelope prov	ease con it back i ided.	splete the to the res	e questions belo earch team at
	As this que place a c use blac leave the	stionnaire will be ross in boxes, k or blue ink to co question or sectio	read by a data implete the form n blank if it doe:	input sca	nner piea: / to you	se :
Section 1	- About You					
Sex	Male 🗌	Female 🗌	Date of birth	11	\square	1
Section 2	- Your Stroke	and the second	alla mia			STATUTE STATUTE
We would	like to know more a	about your stroke	and how it has	impacted	l on your	lite.
What	date did you have yo	ur first stroke?		71	\Box / I	
Have	you had more than or	ne stroke?	Yes	1	No	
If so h	ow many? (Enter nuc	nber in box)				
Thinking a feel and in	bout your life since your daily activities	your stroke, tell u	s about any chi	anges you	a have not	liced in how you
As a result	of your stroke do yo	ou have:		Yes	No	
Any proble	ims walking about?					
Any proble	ms in washing yourse	ell?				
Any proble	ms in dressing yours	e#?				
Any proble	ms in your work life?				Π	
Any proble	ths with carrying out I	housework related	tasks?		$\overline{\Box}$	
	me tokine nart in laier	ure activities?				
Any proble	no taking part in lease			1000		
Any proble Any pain o	r discomfort?					
Any proble Any pain o Any feeting	r discomfort? s of anxiety?					

Research Team

Catherine Homer, Research Fellow, Centre for Health and Social Care Research, Sheffield Hallam University

Dr. Peter Allmark, Principle Lecturer, Centre for Health and Social Care Research, Sheffield Hallam University

Sadiq Bhanbhro, Research Fellow, Centre for Health and Social Care Research, Sheffield Hallam University

Rachel Ibbotson, Research Fellow, Centre for Health and Social Care Research, Sheffield Hallam University

Prof. Angela Tod, Florence Nightingale Foundation Chair of Clinical Nursing Practice Research, School of Nursing, Midwifery and Social Work/Central Manchester NHS Foundation Trust, Faculty of Medical and Human Sciences, University of Manchester